Author's response to reviews

Title: Platinum sensitivity and CD133 expression as risk and prognostic predictors of central nervous system metastases in patients with epithelial ovarian cancer

Authors:

Bo-lin Liu (akingi@sina.com)
Shu-juan Liu (hanliu@fmmu.edu.cn)
Andrius Baskys (abaskys@ucr.edu)
Hong Cheng (chenghong@fmmu.edu.cn)
Ying Han (455355316@qq.com)
Chao Xie (28280228@qq.com)
Hui Song (songhuix@163.com)
Jia Li (598587368@qq.com)
Xiao-yan Xin (neurosurg@126.com)

Version: 2 Date: 11 April 2014

Author's response to reviews: see over
Dear Editor,

On behalf of all authors of this manuscript, I’ve made changes according to the editorial office’s comments.

1. Line numbering:

We’ve revised our manuscript to include line and page numbers.

2. Ethics

We’ve included appropriate ethnical statement in the Methods section (line 118-122, page 5-6).

Thank you for your consideration and please do not hesitate to contact me should you have any further questions.

Yours very sincerely,

Xiaoyan Xin, M.D., Ph.D.
Department of Obstetrics and Gynecology, Xijing Hospital, Fourth Military Medical University, West Changle Road, No.127, Xi’an, Shaanxi Province 710032, People's
Republic of China. Tel: +86 29 84771267; fax: +86 29 84773864; email: neurosurg@126.com